Author's response to reviews

Title: Increased expression of Transcription Factor TFAP2 alpha correlates with chemosensitivity in advanced bladder cancer

Authors:

Iver Nordentoft (Iver.Nordentoft@ki.au.dk)
Lars Dyrskjøt (lars@ki.au.dk)
Julie S Bødker (julie.bodker@ki.au.dk)
Peter J Wild (peter.wild@cell.biol.ethz.ch)
Arndt Hartmann (ardt.hartmann@uk-erlangen.de)
Simone Bertz (Simone.bertz@uk-erlangen.de)
Jan Lehmann (lehmannkiel@arcor.de)
Torben F Ørntoft (orntoft@ki.au.dk)
Karin Birkenkamp-Demtroder (kbdr@ki.au.dk)

Version: 3 Date: 7 April 2011

Author's response to reviews: see over
Dear Editor-in-Chief:

Thank you for accepting our manuscript entitled “Increased expression of Transcription Factor TFAP2α correlates with chemosensitivity in advanced bladder cancer” for publication in BMC Cancer.

We have now corrected the requested changes.

The list of authors in the manuscript is written as they are in the submission system.

The Abstract present in the manuscript matches what is in the submission system.

A short overall title for figure 5 is now provided.

Tables 1 is now included within the text file of the manuscript following the references.

Highlighting and tracking is removed from the manuscript.

The Introduction section is renamed to 'Background'.

The Materials and methods section is renamed to 'Methods'.

Authors' contributions are now formatted into a paragraph.

Address for correspondence
Iver Nordentoft, Department of Molecular Medicine (MOMA) Center for Molecular Clinical Cancer Research (CMCC), Aarhus University Hospital / Skejby, Brendstrupgaardsvej 100, DK-8200 Aarhus N, Denmark.
Email: Iver.Nordentoft@ki.au.dk; Phone: +45 8949 9470, Fax: +45 8949 6018

The authors declare that the manuscript has not been published previously and is not under consideration for publication elsewhere. All of the authors are aware of and agree to the content of the paper and their being listed as an author on the paper.

Best Regards

Iver Nordentoft